Author's response to reviews

Title: Effectiveness and reach of a directed-population approach to improving dental health and reducing inequalities: a cross sectional study

Authors:

Lynn Brewster (lynn.hunter1@nhs.net)
Andrea Sherriff (andrea.sherriff@gla.ac.uk)
Lorna MD Macpherson (Lorna.Macpherson@glasgow.ac.uk)

Version: 2 Date: 19 March 2013

Author's response to reviews:

1. Competing interests of authors have been added
2. Ethics Approval section has been added to the Methods
3. Author’s contributions have been written in format provided.